Dear reviewers and editors,

We have made the following revisions all highlighted in yellow and with track changes, as per your suggestions:

1. We have removed one of the authors, “Patrick Hawkins” due to inability to contact author.
2. We have provided the revised manuscript in a word version.
3. We have added the Institutional review board statement, informed consent statement, and conflict-of-interest statement.
4. We have added a “Core Tip” section.
5. We have addressed each of the subcategories (1-11) in the “COMMENTS” section.

Thank you.